Author’s response to reviews

Title: "Risk factors associated with hypertensive disorders in pregnancy in Nekemte referral Hospital, from July 2015 to June 2017, Ethiopia: case-control study"

Authors:

Leta Dinsa (leta.hinko43@gmail.com)

Almaz Tamene (almaz.tamene@yahoo.com)

Negeso Gejo (negiyeman@gmail.com)

Version: 2 Date: 18 Dec 2019

Author’s response to reviews:

All comments you send me back were corrected very well. Here is the following responds I have made to your comments

1. List of abbreviations were written in the declaration under the title of list of abbreviation and all abbreviation I used are written in texts

2. Duplication of tables and figures. I corrected it and I send you under lists of tables and figures in a separately

3. Cover letter. I deleted it.

4. key words :I have written under my abstract

5. Lastly I send you the clean, nit highlighted and not tracked manuscript Grammar were checked by language expert and all comments your raised were corrected very well.

Thank you